Author’s response to reviews

Title: Association between anxiety and aggression in adolescents: a cross-sectional study

Authors:
Jee Chung (jechung@hanyang.ac.kr)
Gonjin Song (songgonjin@naver.com)
Kitai Kim (kkt0007@daum.net)
Jeong Yee (jjjhello1@naver.com)
Joo Kim (elisekim@ajou.ac.kr)
Kyung Lee (kaylee@cbnu.ac.kr)
Hye sun Gwak (hsgwak@ewha.ac.kr)

Version: 2 Date: 25 Mar 2019

Author’s response to reviews:

Reviewer reports:

A. Bilgic (Reviewer 1): In the revised manuscript, the authors addressed most of the major suggestions raised by Reviewers improving both the main structure and quality of the present paper. I have no further additional comments.

Annabeth Groenman (Reviewer 2): There are still a few minor messy mistakes. For example a dot is missing in the first paragraph and also a reference. Please do one final check for these type of errors.

--> We did final check and corrected mistakes.